Author's response to reviews

Title: A process-based framework to guide NP integration into primary healthcare teams.

Authors:

Damien Contandriopoulos (Damien.contandriopoulos@umontreal.ca)
Astrid Brousselle (Astrid.Brousselle@usherbrooke.ca)
Carl-Ardy Dubois (carl.ardy.dubois@umontreal.ca)
Mélanie Perroux (perrouxmelanie@gmail.com)
Marie-Dominique Beaulieu (marie-dominique.beaulieu@umontreal.ca)
Isabelle Brault (isabelle.brault@umontreal.ca)
Kelley Kilpatrick (kelley.kilpatrick@umontreal.ca)
Danielle D'Amour (danielle.damour@umontreal.ca)
Esther Sansgter-Gormley (egorm@uvic.ca)

Version: 4 Date: 25 November 2014

Author's response to reviews: see over
A process-based framework to guide NP integration into primary healthcare teams

Authors:

Contandriopoulos, Damien (Ph.D), associate professor, Faculty of Nursing, University of Montreal. Damien.contandriopoulos@umontreal.ca

Brousselle, Astrid (Ph.D), full professor, Department of Community Health Sciences, University of Sherbrooke

Dubois, Carl-Ardy (MD, Ph.D), full professor, Faculty of Nursing, University of Montreal

Perroux, Mélanie (M.Sc.), Research officer, University of Montreal Public Health Research Institute

Beaulieu, Marie-Dominique (M.D., FCMF, M.Sc.) full professor, Department of Family Medicine and Emergency Medicine, University of Montreal

Brault, Isabelle, (RN, Ph.D), assistant professor, Faculty of Nursing, University of Montreal

Kilpatrick, Kelley (RN, Ph.D), assistant professor, Faculty of Nursing, University of Montreal

D'Amour, Danielle (RN, Ph.D), full professor, Faculty of Nursing, University of Montreal

Sangster-Gormley, Esther (RN, ARNP, Ph.D), associate professor, School of Nursing, University of Victoria

Acknowledgment

This research was supported by Canadian Institutes of Health Research (Grant number: 238537) and by the Ministry of Health and Social Services of Québec. The authors also want to thank the Fonds de Recherche du Québec - Santé, the Canada Research Chairs program and the Canadian Institutes of Health Research for supporting their research programs.